Author’s response to reviews

Title: Establishing a nationwide emergency department-based syndromic surveillance system for better public health responses in Taiwan

Authors:

Tsung-Shu Joseph Wu (wcsg@msn.com)
Fuh-Yuan Frank Shih (fystone@mail2000.com.tw)
Muh-Yong Yen (muyen1121@gmail.com)
Jiunn-Shyan Julian Wu (julianwu77@hotmail.com)
Shiou-Wen Lu (lsw@cdc.gov.tw)
Chao Hsiung (hsiung@nhri.org.tw)
Kevin Chi-Ming Chang (kevin@cdc.gov.tw)
Jr-How Chou (jchou@cdc.gov.tw)
Yu-Tseng Chu (littleppiiigg@yahoo.com.tw)
Hang Chang (emergency@ms.skh.org.tw)
Chan-Hsien Chiu (zschou@cdc.gov.tw)
Fu-Chiang Richard Tsui (tsui@cbmi.pitt.edu)
Michael M. Wagner (mmw@cbmi.pitt.edu)
Ih-Jen Su (suihjen@nhri.org.tw)
Chwan-Chuen King (cc_king99@hotmail.com)

Version: 10 Date: 17 December 2007

Author’s response to reviews:

Dear Editor,

We’ve revised this version with including ethical approval in page 5 and some minor revised.

Thank you for your kindly review and sincere wishes,

Tsung-Shu Joseph Wu